Author’s response to reviews

Title: Assessing Receptiveness to Change among Primary Healthcare Providers by Adopting the Consolidated Framework for Implementation Research (CFIR)

Authors:

LEE LAN LOW (low.ll@moh.gov.my)
FATHULLAH IQBAL AB RAHIM (fathullah@moh.gov.my)
MOHAMMAD ZABRI JOHARI (mzabri@moh.gov.my)
ZALILAH ABDULLAH (zalilah.a@moh.gov.my)
SITI HAJAR ABDUL AZIZ (sitihajaraziz92@gmail.com)
NUR AJEERAH SUHAIMI (nurajeerah1009@gmail.com)
NORRAFIZAH JAAFAR (norrafizah.j@moh.gov.my)
AINUL NADZIHA MOHD HANAFIAH (ainulnadziha.mh@moh.gov.my)
YUKE LIN KONG (kong.yl@moh.gov.my)
SITI HANIZA MAHMUD (sitihaniza.m@gmail.com)
MOHAMAD ZAIDAN ZULKERPLI (mzaidan.z@moh.gov.my)
KOMATHI PERIALATHAN (komathi.p@moh.gov.my)
NORAZLIN MUHARAM (azlin.m@dosm.gov.my)
NUR HANI ZAINUDIN (nurhanizainuddin@gmail.com)
ZAIKIAH MOHD ZIN (zaikiah@moh.gov.my)
NORAZILAH MOHD ROSLAN (norazilahmr@gmail.com)
TAHIR ARIS (tahir.a@moh.gov.my)
SHAHNAZ MURAD (shahnaz@moh.gov.my)

Version: 3 Date: 27 Jun 2019

Author’s response to reviews:
Authors’ response to Editor comments (3rd round resubmission)

Title: Assessing Receptiveness to Change among Primary Healthcare Providers by Adopting the Consolidated Framework for Implementation Research (CFIR)

Authors:

Low Lee Lan (low.ll@moh.gov.my)
Faithullah Iqbal Ab Rahim (fathullah@moh.gov.my)
Mohammad Zabri Johari (mzabri@moh.gov.my)
Zalilah Abdullah (zalilah.a@moh.gov.my)
Siti Hajar Abdul Aziz (sitihajaraziz92@gmail.com)
Nur Ajeerah Suhaimi (nurajeerah1009@gmail.com)
Norrafizah Jaafar (norrafizah.j@moh.gov.my)
Ainul Nadziha Mohd Hanafiah (ainulnadziha.mh@moh.gov.my)
Kong Yuke Lin (kong.yl@moh.gov.my)
Siti Haniza Mahmud (sitihaniza.m@gmail.com)
Mohamad Zaidan Zulkepli (mzaidan.z@moh.gov.my)
Komathi Perialathan (komathi.p@moh.gov.my)
Norazlin Muharam (azlin.m@dosm.gov.my)
Nur Hani Zainudin (nurhanizainuddin@gmail.com)
Zaikiah Mohd Zin (zaikiah@moh.gov.my)
Norazilah Mohd Roslan (norazilahmr@gmail.com)
Tahir Aris (tahir.a@moh.gov.my)
Shahnaz Murad (shahnazmurad@gmail.com)

Resubmission Version 3: Date: 25th June 2019
Author’s response to Editor Comments:

1. Consent for publication

Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

RESPONSE –

Thank you and noted.

Consent for publication - “Not Applicable”

2. Availability of data and material

In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data and their contact details, for example, an email address.

RESPONSE –

The dataset that support the findings of this article belongs to the Enhanced Primary Healthcare Evaluation study. Request for data can be obtained from Dr. Mohd Azahadi Omar (drazahadi@moh.gov.my), the head of sector for Biostatistics & Data Repository, National Institute of Health, Ministry of Health Malaysia.

3. Role of funder

Please state clearly the role the funder(s) had in your study in the "funding" section of the declarations.

RESPONSE –

This study was made possible with a funding from the National Institutes of Health, Ministry of Health Malaysia research grant [(21)KKM/NIHSEC/800-3/2/1 Jld.5]. The research grant committee provides funding to the study and oversight the project in terms of scientific merit, timely implementation and completion.

4. Clean copy
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

RESPONSE –

Thank you and noted.